University of Waterloo W. H. Cherry

#HL21.1a

[Statistical Highlight #21]

__________________ __________________

ERROR: Sample Error – Why Does Introductory Statistics Teaching Emphasize ‘Random’ Selecting? Section headings to provide a content overview 1. Background I – What is Statistics About?

2. Background II – Key Ideas: Uncertainty, Error and Repetition 3. Why Does Introductory Statistics Teaching Emphasize ‘Random’ Selecting? 4. Appendix 1: Sample Size and Sample Error under EPS

Example HL21.1 5. Appendix 2: Broader Perspectives on the Protocol for Selecting Units 6. Appendix 3: Error Categories, Samples and Populations 7. Appendix 4: Selecting Protocols Beyond EPS

Example HL21.2 8. Appendix 5: Sample Selecting and Unit Inclusion Probabilities 9. Appendix 6: Some Terminology Thirteen tables Five diagrams Two equations

2022-08-20